Author's response to reviews

Title: Is DRE essential for the follow up of prostate cancer patients? A Prospective audit of 194 patients

Authors:

Narasimhan Ragavan (n.ragavan@lancaster.ac.uk)
Vijay K Sangar (vijpol@aol.com)
Sujoy Gupta (sujoyaditi@hotmail.com)
Jennifer Herdmann (jeniffer.herdmann@lthtr.nhs.uk)
Shyam S Matanhelia (shyam.matanhelia@lthtr.nhs.uk)
Michael E Watson (mike.watson@lthtr.nhs.uk)
Rosemary A Blades (rosie.blades@lthtr.nhs.uk)

Version: 3 Date: 14 December 2004

Author's response to reviews:

N Ragavan
Research Fellow in Urology
C/o Mr. Matanhelia's secretary
Royal Preston Hospital
Preston UK
PR2 9HT
Ph 01772 522468
Fax 01772 522946

14-12-04

Dear Sir,

Is DRE essential for the follow up of prostate cancer patients?
A Prospective audit of 194 patients

We thank you for considering our Manuscript for Biomed Central Urology.

We have revised the manuscripts to the format acceptable for BMC.

1. Title page corrected as advised
2. All the highlights removed
3. The references are now in square brackets
4. Authors contribution, conflicts of interest - changed as per advice

I hope these changes would revise the manuscript to be suitable for publication.

Yours sincerely

N Ragavan